Author’s response to reviews

Title: Can the effects of the Mobilization of Vulnerable Elders in Ontario (MOVE ON) implementation be replicated in new settings: An interrupted time series design

Authors:

Julia Moore (MooreJu@smh.ca)

Barbara Liu (Barbara.liu@sunnybrook.ca)

Sobia Khan (sob.khan@mail.utoronto.ca)

Charmalee Harris (harrisch@smh.ca)

Joycelyne Ewusie (ewusieje@mcmaster.ca)

Jemila Hamid (jhamid@mcmaster.ca)

Sharon Straus (sharon.straus@utoronto.ca)

Version: 2 Date: 01 Mar 2019

Author’s response to reviews:

March 1, 2019

RE: Can the effects of the Mobilization of Vulnerable Elders in Ontario (MOVE ON) implementation be replicated in new settings: An interrupted time series design

Ms. No. BGTC-D-18-00317R1

Dear Dr. Aronin and the BMC Geriatrics Editorial Team,

Thank you very much for considering our revised manuscript.
We thank the reviewer for clarifying her comments. Yes, the definition for mobilization in the accuracy study was that used in the audits, specifically, whether the patient was mobilized three times during the 6-hour period of observation [Line 193-195 in manuscript].

We have highlighted the audit method as a limitation in the discussion [Line 374–381 in manuscript].

We have included an updated clean and tracked changed version of our manuscript in our resubmission. Many thanks to the BMC Geriatrics editorial team and our peer reviewers for the valuable feedback they provided on our manuscript. We believe that our paper is aligns greatly with the objectives of Health Services Research are delighted to have the opportunity to resubmit our paper to this journal.

Sincerely,

Sharon E Straus, MD MSc FRCPC
Director, Knowledge Translation Program
Li Ka Shing Knowledge Institute
St. Michael's Hospital
416-864-3068